Author’s response to reviews

Title: Use of glatiramer acetate between 2010-2015: effectiveness, safety and reasons to start GA as first or second line treatment in Swiss multiple sclerosis patients

Authors:

Chiara Zecca (chiara.zecca@eoc.ch)
Giulio Disanto (giulio.disanto@eoc.ch)
Rosaria Sacco (rosaria.sacco@eoc.ch)
Gianna Riccitelli (gianno.Riccitelli@eoc.ch)
Claudio Gobbi (claudio.gobbi@eoc.ch)

Version: 5 Date: 26 Jun 2019

Author’s response to reviews:

Dear Editor,

the declaration "Giulio Disanto is an editorial board member of BMC neurology" has been added in the respective section.

Kind regards

C. Zecca